Author's response to reviews

Title: Is there excess mortality in women screened with mammography? A meta-analysis of non-breast cancer mortality

Authors:

Sylvie Erpeldinger (erpeldinger.sylvie@wanadoo.fr)
Laure Fayolle (fayolle.laure@gmail.com)
Rémy Boussageon (remy.boussageon2@wanadoo.fr)
Marie Flori (mf.flori@medsyn.fr)
Xavier Lainé (xlaine@wanadoo.fr)
Alain Moreau (alainmoreau0917@orange.fr)
François Gueyffier (francois.gueyffier@univ-lyon1.fr)

Version: 3 Date: 10 October 2013

Author's response to reviews:

Version: 2 Date: 20 September 2013
Reviewer: Arnaud Seigneurin
Reviewer's report:
The major compulsory revisions raised in the first version of the manuscript were addressed in this revised manuscript.

Minor Essential Revisions
Page 8, Results section, last paragraph: “The following estimates of excess mortality were obtained after calculating the maximum quantification of screening the negative effect in good-quality methodological studies » This sentence is unclear and could be rephrased.

We wrote this sentence and redesigned the 4 figures.